Author’s response to reviews

Title: Protocol for assessing maternal, environmental and epigenetic risk factors for dental caries in a population of Queensland children

Authors:

Newell Johnson (n.johnson@griffith.edu.au)
Surani Fernando (Surani.fernando@griffith.edu.au)
David Speicher (d.speicher@griffith.edu.au)
Mahmoud Bakr (m.bakr@griffith.edu.au)
Miles Benton (miles.benton84@gmail.com)
Rodney Lea (grccomputationalgenomics@gmail.com)
Paul Scuffham (p.scuffham@griffith.edu.au)
Gabor Mihala (g.mihala@griffith.edu.au)

Version: 1 Date: 30 Jul 2015

Author’s response to reviews:

Ethics: Relevant documents are attached.

Consent: Informed consent for mothers and children are attached as supplementary material.

Availability of supporting data: As the manuscript is a protocol, there are no supporting data.

Author's contribution: Revised as requested